Author's response to reviews

Title: Determinants of neonatal mortality in Nigeria: Evidence from 2008 Demographic and Health Survey.

Authors:

Osita K Ezeh (ezehosita@yahoo.com)
Kingsley E Agho (k.agho@uws.edu.au)
Michael J Dibley (mdibley@health.usyd.edu.com)
John Hall (John.Hall@newcastle.edu.au)
Andrew N Page (a.page@uws.edu.au)

Version: 7
Date: 10 April 2014

Author's response to reviews: see over
BMC Public Health

MANUSCRIPT REVIEW STATUS

MS No.: 1018342708103101

Title of MS: Determinants of neonatal mortality in Nigeria: Evidence from the 2008 Demographic and Health Survey.

Authors’ name: Ezeh, OK, Agho, KE, Dibley MJ, Hall, J and Page, AN

Editor's Comments

Comment 1:

I was disappointed in the conclusions - these were generally vague and either came out of the blue, not from DHS data, or did not actually draw on the data that were available - see below as an example - please ask the authors to strengthen their conclusions.

Response: The conclusion section of the article has been changed as suggested.